Author's response to reviews

Title: Methodological aspects of a GIS-based environmental health inspection program used in the Athens 2004 Olympic and Para Olympic Games

Authors:

Christos Hadjichristodoulou (hadjich@ath.forthnet.gr)
Elpidoforos S Soteriades (esoteria@hsph.harvard.edu)
Virginia Kolonia (hadjich@ath.forthnet.gr)
Matthew E Falagas (matthew.falagas@tufts.edu)
Efstathios Pantelopoulos (hadjich@ath.forthnet.gr)
Georgios Panagakos (hadjich@ath.forthnet.gr)
Varvara Mouchtouri (hadjich@ath.forthnet.gr)
Jeni Kremastinou (hadjich@ath.forthnet.gr)

Version: 7 Date: 23 August 2005

Author's response to reviews:

Dear editor,

We are submitting the final version of the manuscript in which we have corrected the references, we have included in the references the websites and the additional files are referred in the text of the manuscript as you suggested in your last communication

Yours Sincerely

Christos Hadjichristodoulou